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Current Status

* We performed the fault studies

* Due to the booster failure we were able to work whole weekend on the
setting plasma-cascade amplification. We have observed at least 200 increase
in the IR power when going to the very strong focusing in the common
straight (no changes upstream). We want to confirm the result using power

measurement in the certain spectral range.

File ™ “ertical  Timebase

Trigger  Display  Cursors Measure Math  Analysis Lhiilities He

1 —
P1:ampl(C1) P2:ampl(F1) P3:amnl(F1) Pa:ampl(F2)

Me

value 2818 211 mY TA211 my 8777 mv

mea 899.40 mY 7421086 mY 7421086 my 8.77661 mv

rmin 10 my 4211 my T42 1 mY B777 mv

max 26816 4211 mY T4211 my B.777 mv

sdey 468.61 mY

nurn 4 553043 1 1 1

status R v v A

EFTOA[ETACTH 3 imehase -2 msfTrigger [DE]
500 rmiv/7div 500 m/dr 2.00 2.00 mY 100 ms/div | Mormal — 1.50%
-620.0 mv 100 ms/di 100 msfdiv 100 msddiv 25MS 2.5MS/s/Edge  Positive

4553 k 12.637 k# | 2% ¥l= 254672 ms  Ax= B1.2912ms
1 1.6435 % | | m 1 18929 mv||l  -1.5670 mv 2= 1067584 ms 1/4¥= 1230145 Hz
1T 8387 mVip it 23148 mvlr 272058 my

81312020 2:68:52 AM




Near Future Plans

» Verify orbit of the hadrons and electrons in the common section
e Continue work on hadron imprint observation



